Author's response to reviews

Title: Tuberculosis screening of travelers to higher-incidence countries: A cost-effectiveness analysis

Authors:

  Michael Tan (michael_tan325@yahoo.com)
  Dick Menzies (dick.menzies@mcgill.ca)
  Kevin Schwartzman (kevin.schwartzman@mcgill.ca)

Version: 4 Date: 30 May 2008

Author's response to reviews:

Dear Editors,

As requested on May 28, the following formatted changes have been made to our manuscript:

1. The footer, "Traveler tuberculosis screening" has been removed.
2. Author contributions have been replaced with Authors' contributions.
3. All section headings are in sentence case.
4. Titles and numbers have been removed from the image files.
5. The figures have been cropped to minimize white space.

I hope that the changes are satisfactory.

Yours sincerely,

Kevin Schwartzman